\documentstyle[12pt]{amsart}
\newcommand{\BZ}{\Bbb Z} 
\newcommand{\BQ}{\Bbb Q} 
\newcommand{\BR}{\Bbb R} 
\newcommand{\BC}{\Bbb C} 
\title{}
\author{}
\date{}

\setlength{\oddsidemargin}{0.0in}
\setlength{\textwidth}{6.5in}
\setlength{\topmargin}{0.0in}
\setlength{\headheight}{0.0in}
\setlength{\headsep}{0.0in}
\setlength{\textheight}{9.0in}
\pagestyle{empty}

\begin{document}

\maketitle

\begin{center}
{{\Large\bf ALGEBRA I}\\
\bigskip

{\Large\bf HOMEWORK ASSIGNMENT V}\\
\bigskip
{\bf (Turn in underlined problems.)}}
\end{center}

\vspace{.5in}

\noindent
{\bf 
\begin{tabbing}
\underline{\large\bf CHAPTER}\makebox[2in]{}\=\underline{\large\bf PROBLEMS}\\

\bigskip
\noindent
2 \> \underline{2.29}, \underline{2.32}, 2.37,
\underline{2.38}, \underline{2.44}
\end{tabbing}

}
\vspace{.5in}

\noindent
\underline{\large\bf ADDITIONAL PROBLEMS}
\vspace{.25in}

{\bf 

\underline{1.} Show that the lattice of normal
subgroups of a group is modular.


\bigskip

2. Find a modular lattice which is not isomorphic
to the normal subgroup lattice of any group.

\bigskip

3. Find a group whose subgroup lattice is not modular.

}
\end{document}

